Author’s response to reviews

Title: Quality of reporting of systematic reviews and meta-analyses in emergency medicine based on the PRISMA statement

Authors:

Femke Nawijn (f.nawijn-2@umcutrecht.nl; femkenawijn@live.nl)
Wietske Ham (h.w.ham@umcuttrecht.nl)
Roderick Houwert (r.m.houwert@umcutrecht.nl)
Rolf Groenwold (r.h.groenwold@lumc.nl)
Falco Hietbrink (f.hietbrink@umcutrecht.nl)
Diederik Smeeing (d.p.j.smeeing@umcutrecht.nl)

Version: 2 Date: 31 Jan 2019

Author’s response to reviews:

Dear Dr. Prof De Salis,

We would like to thank you for considering this manuscript for publication in BMC Emergency Medicine. We went over all the comments and made changes accordingly. A point-by-point reply to all comment can be found below. We made the adjustments as ask for and hope it is suitable for publication in your journal.

We would be glad to respond to any further questions and comments that you may have.

Kind regards on behalf of all the authors,

Femke Nawijn

-- Please move your tables to after your reference list.
We have moved our tables after the reference list.

--Please ensure affiliations 3 has been numbered correctly. Currently there are two affiliations numbered '2'.

The second 2 was a mistake, we have changed this to 3.

--Please remove your 'Word Count' section from your title page.

We have removed “Word Count”.

-- Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors WH, RH & FH do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

We have provided more clarity on the contributions of each authors since they all qualify for authorship. We have made chances to improve the accuracy of the ‘author’s contributions’ section.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have upload only the final clean version.